RESEARCH ARTICLE

# Sampling is decisive to determination of *Leishmania (Viannia)* species

# Maxy B. De los Santos <sup>1</sup>\*, Steev Loyola<sup>2¤a</sup>, Erika S. Perez-Velez<sup>3</sup>, Rocio del Pilar Santos<sup>4</sup>, Ivonne Melissa Ramírez<sup>5¤b</sup>, Hugo O. Valdivia<sup>1</sup>

 Department of Parasitology, U.S. Naval Medical Research Unit SOUTH (NAMRU SOUTH), Bellavista, Lima, Peru, 2 Facultad de Medicina, Universidad Peruana Cayetano Heredia, Lima, Peru, 3 Departamento Académico de Medicina Humana, Universidad Andina del Cusco, Cusco, Peru, 4 Vysnova, Lima, Peru,
Facultad de Ciencias Biológicas, Universidad Nacional Mayor de San Marcos, Cercado de Lima, Lima, Peru

¤a Current address: Grupo de Investigación UNIMOL, Facultad de Medicina, Universidad de Cartagena, Cartagena de Indias, Colombia

¤b Current address: Faculty of Medicine and Health Sciences, University of Antwerp, Antwerp, Belgium \* maxy.b.delossantos2.ln@health.mil

# Abstract

# Background

Accuracy of molecular tools for the identification of parasites that cause human cutaneous leishmaniasis (CL) could largely depend on the sampling method. Non-invasive or less-invasive sampling methods such as filter paper imprints and cotton swabs are preferred over punch biopsies and lancet scrapings for detection methods of *Leishmania* based on polymerase chain reaction (PCR) because they are painless, simple, and inexpensive, and of benefit to military and civilian patients to ensure timely treatment. However, different types of samples can generate false negatives and there is a clear need to demonstrate which sample is more proper for molecular assays.

# Methodology

Here, we compared the sensitivity of molecular identification of different *Leishmania (Viannia)* species from Peru, using three types of sampling: punch biopsy, filter paper imprint and lancet scraping. Different composite reference standards and latent class models allowed to evaluate the accuracy of the molecular tools. Additionally, a quantitative PCR assessed variations in the results and parasite load in each type of sample.

# **Principal findings**

Different composite reference standards and latent class models determined higher sensitivity when lancet scrapings were used for sampling in the identification and determination of *Leishmania (Viannia)* species through PCR-based assays. This was consistent for genus identification through kinetoplastid DNA-PCR and for the determination of species using FRET probes-based Nested Real-Time PCR. Lack of species identification in some samples correlated with the low intensity of the PCR electrophoretic band, which reflects the low parasite load in samples.



# GOPEN ACCESS

**Citation:** De los Santos MB, Loyola S, Perez-Velez ES, Santos RdP, Ramírez IM, Valdivia HO (2024) Sampling is decisive to determination of *Leishmania (Viannia)* species. PLoS Negl Trop Dis 18(4): e0012113. https://doi.org/10.1371/journal. pntd.0012113

**Editor:** Alain Debrabant, US Food and Drug Administration, UNITED STATES

Received: October 31, 2023

Accepted: March 28, 2024

Published: April 25, 2024

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012113

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> <u>Commons CC0</u> public domain dedication.

Data Availability Statement: All data are within the manuscript and its supporting information. All data is anonymized by codes. The raw data is in S1

Table that permit to replicate the results of our study.

**Funding:** This work was supported by an award from the Armed Forces Health Surveillance Division, Global Emerging Infection Surveillance Branch (AFHSD/GEIS), ProMIS ID 17\_N6\_1.1.3, 2017-2018, to HOV. The funders had no role in data collection and analysis of study design, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Conclusions

The type of clinical sample can directly influence the detection and identification of *Leishmania (Viannia)* species. Here, we demonstrated that lancet scraping samples consistently allowed the identification of more leishmaniasis cases compared to filter paper imprints or biopsies. This procedure is inexpensive, painless, and easy to implement at the point of care and avoids the need for anesthesia, surgery, and hospitalization and therefore could be used in resource limited settings for both military and civilian populations.

# Author summary

Human cutaneous leishmaniasis affects low-income populations living in places far from health services. The early sampling and detection of the parasite are necessary for timely treatment, however there are no uniform sampling criteria, thus the sensitivity of molecular tests may vary due to various factors such as the type of sample, the time of the disease and the parasite load in the lesion.

In this study, we compared the performance of three sampling methods for molecular identification of the genus and *Leishmania (Viannia)* species in Peru. Several analytical methods, including composite reference standards and latent class models, suggested that lancet scraping might be the best approach for parasite genus detection by kDNA-PCR and for parasite species determination by FRET probes-based Nested Real-Time PCR.

# Introduction

The identification of cutaneous leishmaniasis (CL) is based on clinical presentation and direct parasite identification through microscopy or culture, which could be considered the gold standard [1-3]. However, these methods are less sensitive in comparison with DNA-based molecular detection (approximately 60% and 54% of sensitivity for microscopy and culture, respectively) [4-7].

The World Health Organization does not recommend a specific identification method for leishmaniasis nor a type of sample, but emphasizes the importance of using various parasito-logical assays, including microscopy, culture, and molecular techniques, to identify the parasite for timely treatment [8]. In contrast, the Infectious Diseases Society of America (IDSA) and the American Society of Tropical Medicine and Hygiene (ASTMH) recommend using full-thickness skin biopsy specimens for histopathology and cultures and multiple diagnostic approaches to maximize the likelihood of a positive result [9].

Multi Locus Enzyme Electrophoresis (MLEE) has been considered the gold standard for taxonomy and identification of *Leishmania* species [10–12]. However, this technique requires the isolation and mass culture of parasites to obtain sufficient quantities of proteins for enzymatic assays, which is very time-consuming, labor–intensive, expensive and technically demanding [13]. As an alternative method to MLEE, many PCR-based protocols have been developed, these assays only need to collect tissue using different methods and instruments consider invasives such as punch biopsies, syringe aspirate, scalpel, or lancet scraping, and other non-invasives like filter paper impression and rotating a swab, but with variable sensitivities. Molecular assays with higher sensitivity allows not only to characterize the parasite at the genus or species level, but also to quantify the parasite load in the lesion, evaluate the course of

experimental drug therapies, identify virulence factors or drug resistance markers, and support in surveillance and control programs for leishmaniasis [14].

While different PCR-based methods including PCR-RFLP, nested-PCR, real time PCR, PCR-ELISA, oligo chromatography-PCR have been developed to improve the sensitivity of the detection of parasite DNA, few studies have been conducted to assess the effect of the sampling method and site in diagnosis [15–17]. These studies have demonstrated better results when samples are taken from the base and center of the ulcer, and suggested the use of less invasive methods instead of biopsy to avoid local anesthesia, risk of bleeding and infection [18,19]. Furthermore, it has been suggested that the parasite load varies according to the sampling method and site, hence affecting the ability to detect *Leishmania* [20].

It has been suggested that a single smear sample rather than more invasive samples is sufficient to obtain a positive PCR result [21]. Previous reports compared the results obtained by PCR with less sensitive methods, such as microscopy, culture and rapid tests [22–25] as well as the results obtained using different types of PCR [5,26,27]. However, few studies have evaluated the performance of PCR-based molecular tests with different types of clinical samples, invasive and non-invasive, including different physical sample supports such as filter paper and Giemsa-stained slides, with wide variability in sensitivities or percentage of positives, without a clear consensus on the easiest, fastest and most painless collection method for the benefit of military and civilian patients that guarantees accurate identification of the parasites and timely treatment [7,16,28,29].

In the present study, the performance of two molecular assays, kDNA-PCR for genus identification [30] and FRET probes-based Nested Real-Time PCR for species determination [31], were compared using a group of three types of samples, biopsy by punch (Bx), imprint by filter paper (FP) and scraping by lancet (L), each taken from the same patients. Differences in the sensitivity of the identification and determination of *Leishmania (Viannia)* species across the sampling types was evaluated using different models of composite reference standards (CRS) and latent class models (LCM). Likewise, we observed that less intensity of the amplification band in kDNA-PCR were associated with the inability to determine species of the parasite, thus we evaluated the variations in the parasite load that could explain this phenomenon. The molecular results have also been compared with parasite isolation by culture, which was achieved in a subset of patients.

# Methods

#### Ethics statement

This study was approved by the Institutional Review Board of the U.S Naval Medical Research Unit SOUTH (NAMRU SOUTH) in compliance with all applicable federal regulations governing the protection of human subjects (NMRCD.2007.0018).

Trained personnel obtained written informed consent from each enrolled patient and completed a survey to collect demographic, clinical and epidemiological data.

# **Study population**

Clinical samples for this study were selected from a set previously collected between 2013 to 2016 from individuals older than five years-old, non-pregnant in the case of women, who presented with suspected cutaneous leishmaniasis lesions without previous history of treatment in the last six months. All clinical samples included in this study were obtained from the Hospital Militar Central in Lima, and health facilities in Iquitos city (Laboratorio Referencial de Salud Pública, Hospital Iquitos César Garayar García) and Madre de Dios region (Delta 1 Bajo Puquiri, Jorge Chávez Health Center, Nuevo Milenio Health Center, and Laboratorio



**Fig 1. Sampling methods used per patient.** A. Aspirate for culture, B. Scraping by lancet; C. Imprint by Whatman filter paper; D. Biopsy using a 3 mm sterile punch. The photographs were taken by MBD with a Fujifilm FinePix JX250 14 MP Digital Camera.

Referencial de Salud Pública de Madre de Dios). Additionally, blood samples were collected from healthy individuals from non-endemic area as negative control. The different samples obtained were coded and processed blindly, without knowledge of previous or by-pair results from their arrival at the laboratory, to avoid analytical biases during all procedures and stages of this study.

#### **Parasites culture**

After the lesion was disinfected with 70% ethanol and scabs were removed with a gauze, a 21G, 3cc, 1 1/2" syringe with needle containing 0.1 ml of sterile saline solution (1% gentamicin 10 mg/ml, 1% 5-Fluorocytosin 5mg/ml) was inserted to the internal border or the center of the ulcer. Then the syringe was slightly rotated (Fig 1A), and the fluid was aspirated and ejected into a culture tube with USAMRU sterile blood agar. Parasite growth was monitored for two weeks until promastigotes were visible at 400X in an inverted microscope. Cells were cryopreserved in liquid nitrogen for further studies.

#### Lesion sampling

This study selected subjects that had three different types of clinical samples for molecular assays. Subjects with only one or two types of samples were excluded. After aspirating the lesion for parasite isolation, a clinical sample from the internal border of the lesion was collected by scrapping using a sterile lancet (L) which was later stored in a 1.5 ml microtube with absolute ethanol (Fig 1B). Then, a piece of filter paper (FP, Whatman 3 Qualitative) was gently pressed over the lesion to allow the adherence of fluids and lesion material (Fig 1C). The filter paper was dried at room temperature, cut, and stored in a 1.5 ml microtube with absolute ethanol. Finally, local anesthesia with xylocaine at 5% was injected in the internal border of the lesion to take a tissue biopsy (Bx) with a 3 mm punch biopsy instrument (Fig 1D), which was stored in a 1.5 ml microtube with absolute ethanol. All samples were sent at room temperature to NAMRU SOUTH facilities in Lima and they were stored at -20°C until DNA extraction.

# **DNA** extraction

Nucleic acids from the three types of clinical samples (L, FP and Bx) were extracted using the DNeasy Blood & Tissue kit (QIAGEN, Maryland, USA) following the manufacturer's instructions with modifications according to the sample type [32]. Briefly, to obtain optimum DNA yield and quality, we used up to 25 mg of tissue from Bx and L and 1 cm of diameter for FP cut in small pieces. Biological material from L was separated with a 21G sterile needle and minced with the same needle in a microfuge tube. Then, the tube was centrifuged for 5 minutes at 14,000 rpm, supernatant was discarded, and the pellet was dried at 56°C. The material from FP

was extracted by drying the paper on a clean microscopy slide, cut in tiny pieces with a clean blade and heat in 180  $\mu$ L of ATL buffer at 85°C for 10 minutes in a clean 1.5 ml microcentrifuge tube. Finally, tissue obtained by punch biopsy was washed with 0.5 ml of 1X sterile phosphate buffer saline (PBS 1X) to eliminate traces of ethanol, then the tissue was put on a clean microscopy slide and minced with a clean blade. The material obtained was put in a new microcentrifuge tube. All sample types were digested overnight at 56°C with 20  $\mu$ l of Proteinase K in 180  $\mu$ l of ATL buffer. Then, all samples were heated at 70°C for 10 minutes with AL lysis buffer and purified at room temperature by centrifugation at 8,100 rpm through silica membrane DNeasy spin columns. Then, the columns were washed with buffers AW1 an AW2 based in guanidine hydrochloride. Extracted DNA from all the three sample types was eluted by centrifugation (8,100 rpm) at room temperature for 2 minutes with AE buffer (10 mM Tris-Cl, 0.5 mM EDTA; pH 9.0) and then stored at -20°C until use.

# Identification and characterization of Leishmania species

*Leishmania (Viannia)* genus was identified using PCR targeting the minicircle kinetoplast DNA (kDNA-PCR) with specific primers MP1L and MP3H following conditions previously described [30]. *Leishmania* species were detected by FRET probes-based Nested Real-Time PCR based in the identification of single nucleotide polymorphisms (SNP) of the Mannose Phosphate Isomerase (MPI) and 6-Phosphogluconate Dehydrogenase (6PGD) genes using the LightCycler 480 II system (Roche Diagnostics, Basel, Switzerland). LC 480 Genotyping Master and protocol for melting curve analysis with HybProbe format from LC 480 Software v1.5 were used as previously described [31].

# Evaluation of kDNA-PCR amplification band intensity

Variation in 70 bp kDNA-PCR amplification band intensity was assessed by 2.5% agarose gel electrophoresis with GelRed Nucleic Acid gel stain in 1X Tris–Acetate-EDTA buffer (TAE buffer). Invitrogen 100 bp DNA Ladder was used for identification of size amplification band. The amplification bands from different sampling methods were compared with the intensity of the positive control band using *Leishmania (Viannia) braziliensis* LTB300 strain in a Chemi-Doc MP Imaging System with Image Lab software v5.1 (BIO-RAD).

# **Quantitative PCR**

Parasite load was determined on the three paired types of samples for each patient (Bx, FP, L) by quantitative PCR (qPCR) using SYBR Green I Master (Roche). The standard curve to quantify the number of parasites was prepared with 8 dilutions of genomic DNA of *Leishmania* (*Viannia*) braziliensis MHOM/BR/84/LTB300 strain covering a range of  $5 \times 10^{-3}$  to  $5 \times 10^{4}$  parasites based on an equivalence of 83.15 fg to one parasite [33]. A standard curve of endogenous retrovirus 3, ERV3 for human cells covering a range of 160 to 20 x  $10^{3}$  copies was used to normalize [34].

Quantitative PCR was performed in 25  $\mu$ l with 0.2  $\mu$ g/ $\mu$ l DNA sample, 200  $\mu$ M of each primer (MP3H and MP1L for kDNA-qPCR or ERV3-F and ERV3-R for ERV3-qPCR) in a LightCycler 480 II (Roche) as previously described [17,33].

Parasite numbers were normalized in relation to the human endogenous target based on the equation:  $L^{*}10^{6}H = Leishmania$  parasites per  $10^{6}$  human cells = [number of parasites/ (copy number of ERV3/2)]\*  $10^{6}$  as previously described [33].

All procedures described so far are summarized in a flowchart for further understanding (S1 Fig).

#### Tests accuracy estimation

Two approaches were used to mitigate misclassification induced by the use of imperfect tests: composite reference standards (CRS) and latent class models (LCM) [35].

Two CRS were constructed for classifying individuals using the "OR" rule [36]; CRS1, combined results of the kDNA-PCR in the three types of samples and results of the cell culture, and CRS2 combined results of CRS1 and FRET probes-based Nested Real-Time PCR using the three types of samples. Specifically, individuals were classified as *cases* if any of the previously listed assays yielded a positive result. Besides, *non-cases* were individuals who had negative results in all assays.

In LCM, conditional independence was assumed because molecular tests are nucleic acid amplification techniques and the cell culture is based on the parasite's growth and multiplication in a culture medium [37,38]. Based on the observed results of each molecular detection and cell culture, a latent class analysis (LCA) was used to compute the probability of each individual being classified as a *case* or *non-case* [35,39-41]. In the LCA, each test was used as an indicator variable. A total of six models of LCA (LCM) were constructed; three using results from all kDNA-PCR and the cell culture (LCM1-LCM3), and other three using results from all molecular assays (kDNA-PCR and FRET probes-based Nested Real-Time PCR) and cell culture (LCM4-LCM6). As previously described [41-44], basic two-class models were built to classify individuals into two exclusive and exhaustive groups (LCM1, LCM2, LCM4, and LCM5). Parasite load was included as a covariate in LCM2 and LCM5 because it can influence the results of the tests. The use of parasite load (herein: intensity of PCR bands in agarose gel) as a covariate in the model was based on a heuristic approach that described the relationship between high parasite load and a high probability of positive test results [45]. Finally, threeclass models (LCM3 and LCM6) were built as exploratory models, as described elsewhere [42,46], to assess the improvement in model fit.

#### Statistical analysis

Absolute and relative frequencies were used to describe categorical variables. Median (p50) and interquartile range (IQR) were used to summarize numerical variables. Absolute values of parasite load obtained from the qPCR assays were converted to logarithms [function *log*(parasite load)] because the load displayed a non-normal distribution. Then, the parasite load for each paired sample type was plotted individually, and the overall parasite load by sample type and gel electrophoretic band intensity was plotted using box-and-whisker plots. To assess differences in parasite load between paired samples, the Friedman and Dunn's multiple comparison test was used.

LCM1 and LCM2 were implemented using logistic models, while the other ones used the *probit* function due to non-convergence with logistic functions. The rationale for choosing the LCA-derived reference standard was based on parsimony and fit. Specifically, the best LCM was selected using a parsimony criterion (fewer degrees of freedom) and various statistical criteria based on log-likelihood (LL), the Akaike's information criterion (AIC), and the adjusted Bayesian information criterion (aBIC) [41,46]. The response pattern of the best LCM was summarized according to the combination of test results and classes.

The McNemar's chi-square test was used to assess differences in the classification of individuals, and the Cohen's Kappa coefficient was used to assess the agreement between reference standards and assay types in each group of samples. Tests performance was estimated in 2 x 2 contingency tables using the groups and classes defined by the CRS and the best LCM, respectively. The sensitivity and specificity, as well as their respective 95% exact binomial confidence intervals (95%CI), were estimated. A test for receiver operating characteristic (ROC) area equality with Sidak's adjustment for multiple comparisons was also used. Analyses were performed in Stata v17 (StataCorp. 2021. Stata Statistical Software: Release 17. College Station, TX: StataCorp LLC.) and R/Rstudio v2022.07.2 for Windows, and p<0.05 was considered as significant.

# Results

# Characteristics of the study population

From a group of patients previously enrolled between 2013 to 2016 suspicious cutaneous leishmaniasis cases that had at least three different sampling methods were selected (105 patients, 315 samples of Bx, FP and L). Participants came from the Peruvian departments of Amazonas, Ayacucho, Cajamarca, Cusco, Huánuco, Junín, La Libertad, Lima, Loreto, Madre de Dios, San Martín and Ucayali which are endemic areas where we conduct leishmaniasis surveillance. Only three subjects were women, 75 were military and 30 were civilian, with a median age of 29 years-old (Range: 17–78). Most participants (80%) only had a single lesion (Range: 1–8) with a median size of 1.5 cm in diameter (Range: 0.2–15) and only one participant had mucosal lesion with perforation of the nasal septum (Tables 1 and S1. The median time of disease evolution prior to enrollment was two months ranging from one week to eight years, and 10 participants could be classified as chronic cases (more than six months of disease). It was not observed that any clinical trait (Tables 1 and S1) was associated to the molecular results obtained with the three types of samples (p>0.050).

# Four Leishmania (Viannia) species were identified

From the 105 enrolled patients, kDNA-PCR identified *Leishmania (Viannia)* parasites in 85 of them whereas the FRET probes-based Nested Real-Time PCR identified species in 69 of them, most cases were identified as *Leishmania (Viannia) braziliensis* (50 patients, 72.5%) from the departments of Ayacucho, Cusco, Junín, Loreto, Madre de Dios and Ucayali. Additionally, there were three other species identified: *L. (V.) guyanensis* (11 patients, 15.9%) in samples from the departments of Ayacucho, Cusco, Junín, Loreto and San Martín; *L. (V.) lainsoni* in four patients from the departments of Amazonas, Cusco and Loreto (5.9%); one case of *L. (V.) peruviana* from Amazonas (1.4%); and three from Cusco and Loreto showed a putative pattern of *L. (V.) braziliensis/L. (V.) peruviana* hybrids (4.3%) (Fig 2). In 16 patients from Amazonas,

Variable	n (%)
Male	102 (97.1)
Age (years)*	29 (14.0)
1 lesion	84 (80.0)
2 lesions or more	21 (20.0)
Time of disease (months)*	2.0 (3.0)
Lesion size (cm)*	1.5 (1.0)
Enrolled in 2013	16 (15.2)
Enrolled in 2014	27 (25.7)
Enrolled in 2015	37 (35.2)
Enrolled in 2016	25 (23.8)

\*Median (IQR)

https://doi.org/10.1371/journal.pntd.0012113.t001





Cusco, Huánuco, Junín, La Libertad, Lima, Loreto and Ucayali, it was not possible to identify the species of the infecting parasite, which was named *Leishmania* spp.

#### Sample type affected the identification of Leishmania (Viannia)

The analysis of 315 clinical samples (105 patients) showed that the sampling method had an important influence in the identification of *Leishmania (Viannia)*. In terms of genus detection, the kDNA-PCR assay showed 1.9% and 4.8% more positives when imprint by filter paper (FP) and scraping by lancet (L) were used in comparison to biopsy by punch (Bx), (Table 2). Species identification by FRET probes-based Nested Real-Time PCR using L identified 9.5% more positives than FP and 8.6% more positives than Bx (Table 2).

The kDNA-PCR amplification band presented lower intensity in those samples that did not amplify to determine species. The qPCR showed that the *log (parasite load)* of 159 paired samples from 53 individuals, was significantly higher in the FP (p = 0.031) and L samples (p = 0.033) compared to the Bx samples (Table 2). This was also observed in the parasite load of the FP and L samples compared to the Bx samples.

Type of assay/analysis	n (%)						
Cell culture	41 (39.1)						
kDNA-PCR							
Biopsy	79 (75.2)						
Filter paper	81 (77.1)						
Lancet	84 (80.0)						
Nested RT-PCR							
Biopsy	58 (55.2)						
Filter paper	57 (54.3)						
Lancet	67 (63.8)						
qPCR (parasite lo	oad)*						
Biopsy	9.6 (7.3)						
Filter paper	10.0 (5.7)						
Lancet	9.9 (6.5)						
CRS							
CRS1	85 (81.0)						
CRS2	85 (81.0)						

Table 2. Positivity and parasite load by assay and sample type, and composite reference standard (N = 105).

\*Log(Leishmania parasites per 106 human cells)

https://doi.org/10.1371/journal.pntd.0012113.t002

# Higher parasite load was associated with positive results in the FRET probes-based Nested Real-Time PCR

Regardless of the type, if a sample amplified by kDNA-PCR produced a single band with a weak or very weak intensity in agarose gel electrophoresis compared to the positive control (Fig.3A), the likelihood of non-amplification in either or both genes (Mannose Phosphate Isomerase and 6-Phosphogluconate Dehydrogenase) in the FRET probes-based Real-Time PCR increased. Thus, in these samples, the identification of *Leishmania (Viannia)* species was not achieved (Fig.3B) and could correlate with a lower parasite load in comparison to the positive control and other samples with a strong positive band. Similarly, the analysis of the



**Fig 3.** Weak electrophoretic band intensity in kDNA-PCR correlated with the inability to determine species by FRET probes-based Nested Real-Time PCR. A) Electrophoresis in 2.5% agarose gel. Legend: C1, distilled water PCR grade; C2, human DNA as negative control; C3, *L.* (*V.*) *braziliensis* LTB300 DNA as positive control; Bx biopsy samples; FP filter paper samples; L lancet samples. B) Melting peaks of 6PGD gene obtained for FRET probes-based Nested Real-Time PCR. Legend: PAN, *L.* (*V.*) *panamensis*; GUY *L.* (*V.*) *guyanensis*; LAIN, *L.* (*V.*) *lainsoni*; PER, *L.* (*V.*) *peruviana*; BRA, *L.* (*V.*) *braziliensis*.

https://doi.org/10.1371/journal.pntd.0012113.g003



Fig 4. Parasite load is related with intensity of kDNA-PCR electrophoretic band. A) Individual parasite load in paired samples. B) Overall parasite load by sample type. C) Comparative parasite load by type of samples and intensity of electrophoretic bands. Bx: Biopsy; FP: Filter paper imprints; L: Lancet scraping.

parasite load obtained after the qPCR showed differences between the clinical samples, being significantly higher in FP (p = 0.031) and L (p = 0.033) in contrast to Bx (Fig 4B). The same pattern was observed between samples with a strong versus weak or very weak band for kDNA-PCR, where the high parasite load corresponded to the strong bands in all types of samples in contrast to the lowest values of parasite load in the weak or very weak (Fig 4C).

#### Classification of individuals according to CRS and LCM

According to both CRS, the frequency of cases in the study was 81.0% (85/105 cases, Table 2). Individual classification did not differ based on the CRS used (p = 1.000), resulting in a very high degree of agreement between the two CRS (Kappa = 1.000, 95%CI: 1.000–1.000). Based on these findings, subsequent analyzes included only CRS1 because it contained the fewest tests in its construct (Table 2).

LCM1 was identified as the most parsimonious model due to its lower degrees of freedom compared to the other models, whereas LCM6 was the most complex (Table 3). According to the log-likelihood of the model, the AIC and the aBIC, the LCM1 and LCM2 models were the best and allowed to classify 84 (80%) individuals as *cases* (Table 3, Class 2 column). There was no difference in the classification of the individuals (p = 1.000), and both models exhibited high agreement between them (Kappa = 1.000, 95%CI: 1.000–1.000).

Latent Class Models	LL(model)	df	AIC	aBIC	Class 1	Class 2	Class 3
					% (n)		
3 kDNA-PCR + cell culture							
(LCM1) 2-class	-151.99	9	321.98	345.87	20.0 (21)	80.0 (84)	-
(LCM2) 2-class + load	-146.71	10	313.42	339.95	20.0 (21)	80.0 (84)	-
(LCM3) 3-class	-147.53	15	323.07	360.23	20.0 (21)	40.9 (43)	39.1 (41)
3 kDNA-PCR + 3 RT-PCR + cell	l culture						
(LCM4) 2-class	-622.72	15	1275.43	1315.24	32.4 (34)	67.6 (71)	-
(LCM5) 2-class + load	-270.6	16	573.19	615.66	41.9 (44)	58.1 (61)	-
(LCM6) 3-class	-292.83	23	611.66	672.7	19.1 (20)	23.8 (25)	57.1 (60)

Table 3. Different models of Latent Class Analysis (N = 105).

LL(model): log-likelihood of the model. df: degrees of freedom. AIC: Akaike's information criterion. aBIC: Adjusted Bayesian information criterion. LCM: Latent class model. RT-PCR: FRET probes-based Nested Real-Time PCR.

https://doi.org/10.1371/journal.pntd.0012113.t003

kDNA-PCR		Culture	Class 1	Class 2	
Bx	FP	L		n	n
Neg.	Neg.	Neg.	Neg.	20	0
Neg.	Neg.	Pos.	Neg.	1	0
Neg.	Pos.	Pos.	Neg.	0	5
Pos.	Neg.	Pos.	Neg.	0	1
Pos.	Neg.	Pos.	Pos.	0	2
Pos.	Pos.	Neg.	Pos.	0	1
Pos.	Pos.	Pos.	Neg.	0	37
Pos.	Pos.	Pos.	Pos.	0	38

Table 4. Eight different patterns grouped in two classes were generated with the model LCM2.

Bx: Biopsy; FP: Filter paper imprints; L: Lancet scraping; Neg: Negative results; Pos: Positive results

#### https://doi.org/10.1371/journal.pntd.0012113.t004

In contrast to LCM1 (the basic model), LCM2 had a better fit, included parasite load as a covariate, and reproduced the same distribution of individuals by class as LCM1 (Table 3). The LCM2 model generated eight different patterns of results. The first two patterns were grouped in Class 1 with 21 individuals; 20 of which tested negative by kDNA-PCR and cell culture in all samples, while one individual tested positive for the lancet used in the kDNA-PCR (Table 4). The other patterns grouped 84 individuals in Class 2 since they presented at least two positive results in the samples analyzed by kDNA-PCR and one positive or negative result in the cell culture (Table 4).

When comparing CRS1 and LCM2, a nearly identical grouping was observed; 20 individuals classified as *non-cases* by both approaches, 84 individuals as *cases* by both approaches, and one individual was classified as a *non-case* by LCM and as a *case* by CRS. Despite the discrepancy, the standard references were comparable and exhibited a high level of agreement (p = 0.317, Kappa = 0.970, 95%CI: 0.911–1.000). To evaluate the diagnostic performance of each index test, both CRS1 and LCM2 were used in the subsequent analyses.

# CRS and LCM demonstrated no significant differences between samples for our molecular assays

When evaluating the discriminatory capacity of the kDNA-PCR or the FRET probes-based Nested Real-Time PCR through the area under the ROC curve, it did not show significant differences (p>0.050) regardless of the type of sample (Bx, FP, or L) or reference standard evaluated (LCM or CRS). However, in all scenarios where kDNA-PCR were used, there was a significant increase in sensitivity and the area under the ROC curve (p<0.050) when compared to the FRET probes-based Nested Real-Time PCR and cell culture. Although the estimated sensitivity with the LCM2 was higher than the CRS, the difference was not significant (p>0.050). Finally, regardless of the reference standard used (LCM or CRS), the sensitivity of the index tests was higher when lancet scraping samples were used. In general, no significant differences in specificity were found between the evaluated tests (Fig 5).

# Discussion

This study demonstrates that different sampling methods can affect the discrimination capacity of molecular tests for the identification of parasite and determination of *Leishmania (Viannia)* species. Likewise, the ability to detect the parasite by PCR amplification is associated with the parasite load in the samples. These observations were independent from clinical



**Fig 5. Sensitivity and specificity for each reference and technique in relation to the type of biological sample.** LCM: Latent class model; CRS: Composite reference standard; RT-PCR: FRET probes-based Nested Real-Time PCR; L: Lancet scraping; FP: Filter paper imprints; Bx: Biopsy.

parameters as age of patients, time of disease evolution (considering <6 months for acute cutaneous leishmaniasis and  $\geq$ 6 months for chronic cutaneous leishmaniasis, like [47]), size and location of lesions, number of lesions, and the *Leishmania* species identified in this study.

Similar to our previous report [31], the molecular assays allowed us to identify the species of *Leishmania (Viannia)* in most clinical samples. Interestingly, despite of not being the objective of the study, our results confirmed the wide geographical distribution of *Leishmania* species observed in previous reports [48,49], possibly due to an increase in human migration and environmental changes observed in the recent years [50]. Similar to previous studies, *L. (V.) braziliensis* was the most prevalent species followed by *L. (V.) guyanensis* and *L. (V.) lainsoni* [48,51,52].

The detection of the genus *Leishmania* by kDNA-PCR, although it does not show significant differences between sample types, does not mean that there were no differences between the sensitivities obtained for each type of sample, with the lancet being similar to filter paper and superior to biopsies. This could be explained by the greater amount of parasite DNA in the superficial layers of the skin taken with the lancet and filter paper (72) unlike a biopsy that can collect a greater amount of skin tissue and therefore human DNA (15). Moreover, the need of local anesthesia and surgery with pain and distress for patients become impractical the biopsy in field conditions [17]. Therefore, we encourage the use of non-invasive and outpatient methods such as lancet scraping or filter paper impressions, with results similar to invasive methods such as biopsy, with the advantage of avoiding distress, pain and in some cases hospitalization in civilian and military patients.

Previous studies using scrapping employed a number 11 sterile scalpel or a number 15 surgical blade [5,7,15], but we prefer the use of a sterile aluminum lancet given the possibility of bleeding due to mishandling of the scalpel. Another advantage of lancet scraping was obtaining sufficient material for our molecular assays rather than invasive samples (aspirates, biopsies), similar to the results of a meta-analysis suggesting that smears are very sensitive tools for the detection of *Leishmania* instead of invasive biopsies or aspirates [21]. Additionally, the smear obtained by lancets can be used in microscopy [5] and put in filter paper for DNA extraction and detection by PCR [53].

Regarding filter paper, we prefer to work with Whatman 3 Qualitative because its coarse porosity matches that of previous studies [4,5,7,53,54]. A simple and soft pressure on the lesion

allowed obtaining enough tissue fluid and recovering DNA from the parasites without the need to put biopsy fragments or smears on the filter paper [53,54].

Noteworthy, for more than 20 years studies have suggested the use of scraping with lancet or scalpel as an alternative to biopsies for the diagnosis of cutaneous leishmaniasis by PCR [16,55] to avoid the need of personnel specialized in surgery and iatrogenic bacterial infections [16,56]. Despite this evidence, the systematic use of biopsies continued, especially for comparison of PCR with microscopy [57,58] or to study the torpid evolution of the disease [59]. Furthermore, biopsies stored in formalin require trained personnel in histology to identify the parasite, but the determination of species is not possible on this type of sample.

Another alternative non-invasive sampling methods like swabs, tape strip disc and cytology brushes may be particularly useful, especially for lesions located on mucosal areas and in body zones with higher sensitivity [18,60,61] or in the case of children [62–64]. Similar to lancets and filter papers, these methods reduce the risk of bleeding and exposure of health personnel to body fluids [4,61,65].

There is a need for a standardized sampling system in function of the clinical characteristics of the patients. For example, low parasite load of *Leishmania (Viannia) braziliensis* is observed in the chronic cutaneous disease (lesions persist for more than six months) [66], but the criteria to establish a division between acute and chronic disease are missing, since more than three months and 12 months were previously was considered chronic cutaneous leishmaniasis [4,7,15,17].

The high sensitivity and apparent homogeneity from kDNA-PCR to detect the parasite, independently from the type of samples used, could be explained by the large number of copies of the minicircle fragment in the kinetoplastid DNA to be amplified and the interspecific constitutive homogeneity of the amplified minicircle sequence [67]. Other targets have been analyzed principally in Old World *Leishmania* species through PCR-RFLP, DNA sequencing and melting analysis, but the sensitivity is highly variable [68].

Melting curve analysis by Real-Time PCR has been used to identify *Leishmania* species in the Old World [54,69,70]. In a previous work, we demonstrated a sensitivity of 92% and a specificity of 77% of FRET probes-based Nested Real-Time PCR which was much higher than identification by culture, microscopy and leishmanin skin test in Peruvian clinical samples [31]. The current study showed no significant differences between types of samples compared, and like kDNA-PCR, when the punctual estimates of the sensitivities are taken, this molecular tool shows higher values when the lancet is used in contrast to filter paper and biopsy.

A limitation in all the tests to identify the parasite, even by molecular methods, is the presence of false negatives. This study demonstrated an association between false negatives for the determination of parasite species by FRET probes-based Nested Real-Time PCR and the reduction in the intensity of the amplification band by kDNA-PCR as described in a previous study that demonstrates that it is not possible to determine species using different targets by PCR when the kDNA-PCR results show a low intensity of the amplified band [20]. Another limitation of this method for species determination is the amount of amplified target DNA, basically only two gene copies present in the parasite nucleus for MPI and 6PGD, in contrast to the thousands of copies of the kinetoplast region that is amplified in kDNA-PCR [71] which confers higher sensitivity to the molecular detection of the parasite, regardless of the type of sample.

Unidentified species were associated with low parasite load independent of the sample type, similar to previous reports [17,33]. The results underline the utility of less invasive samples (L and FP versus Bx) for PCR diagnosis which is supported by a report that demonstrates that parasite load in superficial tissue is higher than inferior dermis and hypodermis [72].

The start of antileishmanial treatment does not necessarily depend on the knowledge of the infecting *Leishmania* species. However, it has been observed that certain species can influence treatment outcome [73,74], its failure [75] and the choice of treatment [9]. Furthermore, species identification has been highly recommended as part of clinical trials [76] and study of cutaneous leishmaniasis surveillance and outbreaks [77].

Similarly, the precision and reproducibility of the different sampling methods that guarantee a good performance of our molecular tests, has been key to the identification of *Leishmania* (*Viannia*) species in groups of sandflies described as putative vectors of the disease in the department of Madre de Dios, Peru [78–80]. Likewise, our study of population genetics and distribution of *L.* (*V.*) *braziliensis* in Peru was possible after the identification of the parasites [81]. Other studies where we evaluated two different types of rapid tests took advantage from the identification of *Leishmania* species for inclusion criteria such as one study that tested RPA-LF [23,82] and another on the CL-Detect device [83] both to be applied under field conditions. It should be noted that an additional advantage of our procedure is that the samples can be included in absolute ethanol at room temperature and thus be sent without the need for cold chain from the sampling sites to the laboratory where the molecular assays are carried out. Our findings support the use of minimal invasive sampling methods, such as scraping with lancets or filter paper impressions as alternatives to guarantee the correct identification of the parasite through molecular tests and thus ensure a more standardized management of the parasites mainly under field conditions.

# Conclusions

This study shows that the clinical sample collection procedure can directly influence the sensitivity of the FRET probe-based Nested Real-Time PCR that we developed to identify *Leishmania (Viannia)* species. In this sense, samples collected by scraping with lancets could be just as appropriate to identify the parasite in a similar way to filter paper and as an alternative to invasive methods such as biopsy. Standardized use of minimally invasive sampling methods is preferred for the benefit of the patient and potentially applicable to other molecular identification assays under field conditions.

# Supporting information

**S1 Fig. Flowchart that summarize all experimental procedures of this study.** (TIF)

**S1 Table. Clinical data of study population.** All data were anonymized by coding and processed blindly as mentioned above. Legend for columns B to I: 1 positive result, 0 negative result, \* weak/very weak amplification band in kDNA-PCR electrophoresis. Legend for column S, work: FFAA, Peruvian armed forces. Legend for column U, *Leishmania (Viannia)* species: BRA, *L. (V.) braziliensis*; GUY, *L. (V.) guyanensis*; LAIN, *L. (V.) lainsoni*; PER, *L. (V.) peruviana*; spp, *Leishmania* spp.; PER-BRA, possible hybrid *L. (V.) peruviana-L. (V.) braziliensis*; Neg, negative. The raw data in this table permit replicate the results of our study. (XLSX)

# Acknowledgments

The authors would like to thank Edson Maguiña, Luis Angel Rosales and Dolores Rimarachín for their support in sample collection. Also, we want to thank Carmen Lucas for her invaluable administrative assistance. Finally, the authors are grateful to all patients who participated voluntarily in this study.

#### Disclaimer

Some authors of this manuscript are employees of the U.S. Government. This work was prepared as part of their official duties. Title 17 U.S.C. \$105 provides that "Copyright protection under this Title is not available for any work of the United States Government". Title 17 U.S.C. \$101 defines a U.S. Government work as a work prepared by a military service member or employee of the U.S. Government as part of that person's official duties.

The views expressed in this article are those of the author and do not necessarily reflect the official policy or position of the Department of the Navy, Department of Defense, nor the U.S. Government.

# **Author Contributions**

- **Conceptualization:** Maxy B. De los Santos, Erika S. Perez-Velez, Rocio del Pilar Santos, Ivonne Melissa Ramírez.
- **Data curation:** Maxy B. De los Santos, Steev Loyola, Erika S. Perez-Velez, Rocio del Pilar Santos, Ivonne Melissa Ramírez, Hugo O. Valdivia.
- Formal analysis: Maxy B. De los Santos, Steev Loyola, Erika S. Perez-Velez, Rocio del Pilar Santos, Ivonne Melissa Ramírez.
- Funding acquisition: Hugo O. Valdivia.
- Investigation: Maxy B. De los Santos, Steev Loyola, Rocio del Pilar Santos, Ivonne Melissa Ramírez.
- Methodology: Maxy B. De los Santos, Steev Loyola, Erika S. Perez-Velez, Rocio del Pilar Santos, Ivonne Melissa Ramírez.
- Project administration: Hugo O. Valdivia.
- Resources: Maxy B. De los Santos.
- Software: Steev Loyola, Ivonne Melissa Ramírez.
- **Supervision:** Maxy B. De los Santos, Erika S. Perez-Velez, Rocio del Pilar Santos, Hugo O. Valdivia.
- Validation: Maxy B. De los Santos, Steev Loyola, Erika S. Perez-Velez, Rocio del Pilar Santos, Ivonne Melissa Ramírez.
- Visualization: Maxy B. De los Santos, Steev Loyola, Erika S. Perez-Velez, Rocio del Pilar Santos, Hugo O. Valdivia.
- Writing original draft: Maxy B. De los Santos, Rocio del Pilar Santos.
- Writing review & editing: Maxy B. De los Santos, Steev Loyola, Erika S. Perez-Velez, Rocio del Pilar Santos, Ivonne Melissa Ramírez, Hugo O. Valdivia.

#### References

- Douba MD, Abbas O, Wali A, Nassany J, Aouf A, Tibbi MS, et al. Chronic cutaneous leishmaniasis, a great mimicker with various clinical presentations: 12 years experience from Aleppo. J Eur Acad Dermatology Venereol. 2012; 26: 1224–1229. https://doi.org/10.1111/j.1468-3083.2011.04266.x PMID: 21958339
- Reithinger R, Dujardin JC, Louzir H, Pirmez C, Alexander B, Brooker S. Cutaneous leishmaniasis. Lancet Infectious Diseases. 2007. pp. 581–596. https://doi.org/10.1016/S1473-3099(07)70209-8 PMID: 17714672

- de Vries HJC, Reedijk SH, Schallig HDFH. Cutaneous Leishmaniasis: Recent Developments in Diagnosis and Management. American Journal of Clinical Dermatology. Springer International Publishing; 2015. pp. 99–109. https://doi.org/10.1007/s40257-015-0114-z PMID: 25687688
- Boggild AK, Valencia BM, Espinosa D, Veland N, Ramos AP, Arevalo J, et al. Detection and Species Identification of *Leishmania* DNA from Filter Paper Lesion Impressions for Patients with American Cutaneous Leishmaniasis. Clin Infect Dis. 2010; 50: e1–e6. <u>https://doi.org/10.1086/648730</u> PMID: 19947858
- Bensoussan E, Nasereddin A, Jonas F, Schnur LF, Jaffe CL. Comparison of PCR assays for diagnosis of cutaneous leishmaniasis. J Clin Microbiol. 2006; 44: 1435–1439. <u>https://doi.org/10.1128/JCM.44.4.</u> 1435-1439.2006 PMID: 16597873
- Faber WR, Oskam L, van Gool T, Kroon NCM, Knegt-Junk KJ, Hofwegen H, et al. Value of diagnostic techniques for cutaneous leishmaniasis. J Am Acad Dermatol. 2003; 49: 70–74. <u>https://doi.org/10.1067/mjd.2003.492</u> PMID: 12833011
- Eroglu F, Uzun S, Koltas IS. Comparison of clinical samples and methods in chronic cutaneous leishmaniasis. Am J Trop Med Hyg. 2014; 91: 895–900. <u>https://doi.org/10.4269/ajtmh.13-0582</u> PMID: 25223940
- 8. WHO. Control of the leishmaniasis: report of a meeting of the WHO Expert Committee on the Control of Leishmaniases. Geneva; 2010.
- Aronson N, Herwaldt BL, Libman M, Pearson R, Lopez-Velez R, Weina P, et al. Diagnosis and treatment of leishmaniasis: Clinical practice guidelines by the infectious diseases society of America (IDSA) and the American Society of tropical medicine and hygiene (ASTMH). American Journal of Tropical Medicine and Hygiene. American Society of Tropical Medicine and Hygiene; 2017. pp. 24–45. https:// doi.org/10.4269/ajtmh.16-84256 PMID: 27927991
- Rioux JA, Lanotte G, Serres E, Pratlong F, Bastien P, Perieres J. Taxonomy of *Leishmania*. Use of isoenzymes. Suggestions for a new classification. Annales de Parasitologie Humaine et Comparee. 1990. pp. 111–125. https://doi.org/10.1051/parasite/1990653111 PMID: 2080829
- Dujardin JC, Bañuls AL, Dujardin JP, Arevalo J, Tibayrenc M, Le Ray D. Comparison of chromosome and isoenzyme polymorphism in geographical populations of *Leishmania (Viannia) peruviana*. Parasitology. 1998; 117: 547–554. https://doi.org/10.1017/S0031182098003357 PMID: 9881379
- Cupolillo E, Brahim LR, Toaldo CB, Paes de Oliveira-Neto M, De Brito MEF, Falqueto A, et al. Genetic polymorphism and molecular epidemiology of *Leishmania (Viannia) braziliensis* from different hosts and geographic areas in Brazil. J Clin Microbiol. 2003; 41: 3126–3132. <u>https://doi.org/10.1128/JCM.41.</u> 7.3126–3132.2003
- Bañuls AL, Hide M, Prugnolle F. Leishmania and the Leishmaniases: A Parasite Genetic Update and Advances in Taxonomy, Epidemiology and Pathogenicity in Humans. Adv Parasitol. 2007; 64. <u>https:// doi.org/10.1016/S0065-308X(06)64001-3</u> PMID: 17499100
- Reithinger R, Dujardin JC. Molecular diagnosis of leishmaniasis: Current status and future applications. Journal of Clinical Microbiology. 2007. pp. 21–25. https://doi.org/10.1128/JCM.02029-06 PMID: 17093038
- Ramirez JR, Agudelo S, Muskus C, Alzate JF, Berberich C, Barker D, et al. Diagnosis of cutaneous leishmaniasis in Colombia: The sampling site within lesions influences the sensitivity of parasitologic diagnosis. J Clin Microbiol. 2000; 38: 3768–3773. https://doi.org/10.1128/JCM.38.10.3768-3773.2000 PMID: 11015400
- Matsumoto T, Hashiguchi Y, Gomez EA, Calvopiña MH, Nonaka S, Saya H, et al. Comparison of PCR results using scrape/exudate, syringe-sucked fluid and biopsy samples for diagnosis of cutaneous leishmaniasis in Ecuador. Trans R Soc Trop Med Hyg. 1999; 93: 606–7. https://doi.org/10.1016/s0035-9203 (99)90065-2 PMID: 10717745
- Suárez M, Valencia BM, Jara M, Alba M, Boggild AK, Dujardin JC, et al. Quantification of *Leishmania* (*Viannia*) kinetoplast DNA in ulcers of cutaneous leishmaniasis reveals inter-site and intersampling variability in parasite load. PLoS Negl Trop Dis. 2015; 9: 1–14. https://doi.org/10.1371/journal.pntd. 0003936 PMID: 26204525
- Taslimi Y, Sadeghipour P, Habibzadeh S, Mashayekhi V, Mortazavi H, Müller I, et al. A novel non-invasive diagnostic sampling technique for cutaneous leishmaniasis. PLoS Negl Trop Dis. 2017; 11. <a href="https://doi.org/10.1371/journal.pntd.0005750">https://doi.org/10.1371/journal.pntd.0005750</a> PMID: 28704463
- Edileuza Felinto de Brito M, Sandra Andrade M, Lima de Almeida E, Angela Cristina Rapela Medeiros, Pereira Werkhäuser R, Isabele Freitas de Ará ujo A, et al. Occupationally Acquired American Cutaneous Leishmaniasis. Case Rep Dermatol Med. 2012; 2012. https://doi.org/10.1155/2012/279517 PMID: 23227369
- 20. Veland N, Boggild AK, Valencia C, Valencia BM, Llanos-Cuentas A, Van Der Auwera G, et al. *Leishmania (Viannia)* species identification on clinical samples from cutaneous leishmaniasis patients in

Peru: Assessment of a molecular stepwise approach. J Clin Microbiol. 2012; 50: 495–498. https://doi.org/10.1128/JCM.05061-11 PMID: 22116151

- Mesa LE, Manrique R, Muskus C, Robledo SM. Test accuracy of polymerase chain reaction methods against conventional diagnostic techniques for Cutaneous Leishmaniasis (CL) in patients with clinical or epidemiological suspicion of CL: Systematic review and meta-analysis. PLoS Negl Trop Dis. 2020; 14: e0007981. https://doi.org/10.1371/journal.pntd.0007981 PMID: 31961871
- 22. Pourmohammadi B, Motazedian M, Hatam G, Kalantari M, Habibi P, Sarkari B. Comparison of three methods for diagnosis of cutaneous leishmaniasis. Iran J Parasitol. 2010. PMID: 22347259
- Saldarriaga OA, Castellanos-Gonzalez A, Porrozzi R, Baldeviano GC, Lescano AG, de Los Santos MB, et al. An Innovative Field-Applicable Molecular Test to Diagnose Cutaneous *Leishmania Viannia* spp. Infections. PLoS Negl Trop Dis. 2016. https://doi.org/10.1371/journal.pntd.0004638 PMID: 27115155
- Tiwananthagorn S, Kato H, Yeewa R, Muengpan A, Polseela R, Leelayoova S. Comparison of LAMP and PCR for molecular mass screening of sand flies for *Leishmania martiniquensis* infection. Mem Inst Oswaldo Cruz. 2017. https://doi.org/10.1590/0074-02760160254 PMID: 28177044
- 25. Bennis I, Verdonck K, Riyad M, El Khalfaoui N, Fellah H, Sahibi H, et al. Accuracy of a rapid test for the diagnosis of cutaneous leishmaniasis in patients with suggestive skin lesions in Morocco. Trop Med Int Heal. 2017. https://doi.org/10.1111/(ISSN)1365-3156
- Cruz I, Millet A, Carrillo E, Chenik M, Salotra P, Verma S, et al. An approach for interlaboratory comparison of conventional and real-time PCR assays for diagnosis of human leishmaniasis. Exp Parasitol. 2013. https://doi.org/10.1016/j.exppara.2013.03.026 PMID: 23562705
- Mouttaki T, Morales-Yuste M, Merino-Espinosa G, Chiheb S, Fellah H, Martin-Sanchez J, et al. Molecular diagnosis of cutaneous leishmaniasis and identification of the causative *Leishmania* species in Morocco by using three PCR-based assays. Parasites and Vectors. 2014; 7. <a href="https://doi.org/10.1186/1756-3305-7-420">https://doi.org/10.1186/1756-3305-7-420</a> PMID: 25189460
- Adams ER, Gomez MA, Scheske L, Rios R, Marquez R, Cossio A, et al. Sensitive diagnosis of cutaneous leishmaniasis by lesion swab sampling coupled to qPCR. <u>https://doi.org/10.1017/</u> S0031182014001280 PMID: 25111885
- Al-Jawabreh A, Dumaidi K, Ereqat S, Nasereddin A, Azmi K, Al-Jawabreh H, et al. A comparison of the efficiency of three sampling methods for use in the molecular and conventional diagnosis of cutaneous leishmaniasis. Acta Trop. 2018. https://doi.org/10.1016/j.actatropica.2018.03.001 PMID: 29522706
- Lopez M, Inga R, Cangalaya M, Echevarria J, Llanos-Cuentas A, Orrego C, et al. Diagnosis of *Leishmania* using the polymerase chain reaction: A simplified procedure for field work. Am J Trop Med Hyg. 1993; 49: 348–356. https://doi.org/10.4269/ajtmh.1993.49.348 PMID: 8396860
- Tsukayama P, Núñez JH, De Los Santos M, Soberón V, Lucas CM, Matlashewski G, et al. A FRET-Based Real-Time PCR Assay to Identify the Main Causal Agents of New World Tegumentary Leishmaniasis. PLoS Negl Trop Dis. 2013; 7: e1956. <u>https://doi.org/10.1371/journal.pntd.0001956</u> PMID: 23301111
- 32. DNeasy ® Blood & Tissue Handbook W W W. Q I A G E N. C O M. 2006.
- Jara M, Adaui V, Valencia BM, Martinez D, Alba M, Castrillon C, et al. Real-time PCR assay for detection and quantification of *Leishmania (Viannia)* organisms in skin and mucosal lesions: Exploratory study of parasite load and clinical parameters. J Clin Microbiol. 2013; 51: 1826–1833. https://doi.org/10.1128/JCM.00208-13 PMID: 23554201
- Yuan CC, Miley W, Waters D. A quantification of human cells using an ERV-3 real time PCR assay. J Virol Methods. 2001; 91: 109–17. https://doi.org/10.1016/s0166-0934(00)00244-5 PMID: 11164492
- Dendukuri N, Schiller I, De Groot J, Libman M, Moons K, Reitsma J, et al. Concerns about composite reference standards in diagnostic research. BMJ. 2018; 360. https://doi.org/10.1136/bmj.j5779 PMID: 29348126
- Reitsma JB, Rutjes AWS, Khan KS, Coomarasamy A, Bossuyt PM. A review of solutions for diagnostic accuracy studies with an imperfect or missing reference standard. Journal of Clinical Epidemiology. 2009. https://doi.org/10.1016/i.iclinepi.2009.02.005 PMID: 19447581
- Koukounari A, Jamil H, Erosheva E, Shiff C, Moustaki I. Latent class analysis: Insights about design and analysis of schistosomiasis diagnostic studies. PLoS Negl Trop Dis. 2021; 15. <u>https://doi.org/10. 1371/journal.pntd.0009042</u> PMID: 33539357
- 38. Chaouch M, Aoun K, Othman S Ben, Abid M Ben, Sghaier I Ben, Bouratbine A, et al. Development and assessment of *Leishmania major* and *Leishmania tropica* specific loop-mediated isothermal amplification assays for the diagnosis of cutaneous leishmaniasis in Tunisia. Am J Trop Med Hyg. 2019; 101. https://doi.org/10.4269/ajtmh.19-0097 PMID: 31094311
- Salomón OD, Pérez AA, Riarte AR, Casas N, Fragueiro-Frías V, Negri V, et al. Performance of rapid tests for canine visceral leishmaniasis diagnosis in Argentina. Medicina (B Aires). 2020; 80.

- 40. Cota GF, De Sousa MR, De Freitas Nogueira BM, Gomes LI, Oliveira E, Assis TSM, et al. Comparison of parasitological, serological, and molecular tests for visceral leishmaniasis in HIV-infected patients: A cross-sectional delayed-type study. Am J Trop Med Hyg. 2013; 89. https://doi.org/10.4269/ajtmh.13-0239 PMID: 23836568
- 41. Fraga DBM, Pacheco LV, Borja LS, Tuy PG da SE, Bastos LA, Solcà M da S, et al. The Rapid Test Based on *Leishmania infantum* Chimeric rK28 Protein Improves the Diagnosis of Canine Visceral Leishmaniasis by Reducing the Detection of False-Positive Dogs. PLoS Negl Trop Dis. 2016; 10. https://doi. org/10.1371/journal.pntd.0004333 PMID: 26731098
- Boelaert M, Aoun K, Liinev J, Goetghebeur E, Van Der Stuyft P. The potential of Latent Class Analysis in diagnostic test validation for canine *Leishmania infantum* infection. Epidemiol Infect. 1999; 123. https://doi.org/10.1017/S0950268899003040 PMID: 10694163
- **43.** Boelaert M, Rijal S, Regmi S, Singh R, Karki B, Jacquet D, et al. A comparative study of the effectiveness of diagnostic tests for visceral leishmaniasis. Am J Trop Med Hyg. 2004; 70. <u>https://doi.org/10.</u> 4269/ajtmh.2004.70.72 PMID: 14971701
- 44. Keter AK, Lynen L, Van Heerden A, Goetghebeur E, Jacobs BKM. Implications of covariate induced test dependence on the diagnostic accuracy of latent class analysis in pulmonary tuberculosis. J Clin Tuberc Other Mycobact Dis. 2022; 29. https://doi.org/10.1016/j.jctube.2022.100331 PMID: 36111071
- 45. Schumacher SG, van Smeden M, Dendukuri N, Joseph L, Nicol MP, Pai M ZH. Diagnostic test accuracy in childhood pulmonary tuberculosis: a bayesian latent class analysis. Am J Epidemiol. 2016; 184: 690– 700. https://doi.org/10.1093/aje/kww094 PMID: 27737841
- 46. Wiegand RE, Cooley G, Goodhew B, Banniettis N, Kohlhoff S, Gwyn S, et al. Latent class modeling to compare testing platforms for detection of antibodies against the *Chlamydia trachomatis* antigen Pgp3. Sci Rep. 2018; 8. https://doi.org/10.1038/s41598-018-22708-9 PMID: 29523810
- Weigle KA, Labrada LA, Lozano C, Santrich C, Barker DC. PCR-based diagnosis of acute and chronic cutaneous leishmaniasis caused by *Leishmania (Viannia)*. J Clin Microbiol. 2002; 40: 601–606. <u>https:// doi.org/10.1128/JCM.40.2.601–606.2002</u>
- Lucas CM, Franke ED, Cachay MI, Tejada A, Cruz ME, Kreutzer RD, et al. Geographic distribution and clinical description of leishmaniasis cases in Peru. Am J Trop Med Hyg. 1998; 59: 312–317. <u>https://doi.org/10.4269/ajtmh.1998.59.312</u> PMID: 9715953
- 49. Kato H, Cáceres AG, Seki C, García CRS, Mauricci CH, Martínez SCC, et al. Further insight into the geographic distribution of *Leishmania* species in Peru by cytochrome b and mannose phosphate isomerase gene analyses. PLoS Negl Trop Dis. 2019;13. <u>https://doi.org/10.1371/journal.pntd.0007496</u> PMID: 31220120
- Berry I, Berrang-Ford L. Leishmaniasis, conflict, and political terror: A spatio-temporal analysis. Soc Sci Med. 2016; 167: 140–149. https://doi.org/10.1016/j.socscimed.2016.04.038 PMID: 27194448
- Lucas CM, Franke ED, Cachay MI, Tejada A, Carrizales D, Kreutzer RD. Leishmania (Viannia) lainsoni: first isolation in Peru. Am J Trop Med Hyg. 1994; 51: 533–7. Available: <u>http://www.ncbi.nlm.nih.gov/</u> pubmed/7985744
- 52. Kato H, Cáceres AG, Seki C, Silupu García CR, Holguín Mauricci C, Castro Martínez SC, et al. Further insight into the geographic distribution of *Leishmania* species in Peru by cytochrome b and mannose phosphate isomerase gene analyses. Brodskyn CI, editor. PLoS Negl Trop Dis. 2019; 13: e0007496. https://doi.org/10.1371/journal.pntd.0007496 PMID: 31220120
- Mota CA, Venazzi EAS, Zanzarini PD, Aristides SMA, Lonardoni MVC, Silveira TGV. Filter paper performance in PCR for cutaneous leishmaniasis diagnosis. Rev Soc Bras Med Trop. 2021; 54. <u>https://doi.org/10.1590/0037-8682-0047-2020</u> PMID: 33338104
- de Monbrison F, Mihoubi I, Picot S. Real-time PCR assay for the identification of cutaneous Leishmania parasite species in Constantine region of Algeria. Acta Trop. 2007; 102: 79–83. <u>https://doi.org/10.1016/j.actatropica.2007.04.001 PMID: 17485067</u>
- 55. Belli A, Rodriguez B, Aviles H, Harris E. Simplified polymerase chain reaction detection of New World Leishmania in clinical specimens of cutaneous Leishmaniasis. Am J Trop Med Hyg. 1998. <u>https://doi.org/10.4269/ajtmh.1998.58.102</u> PMID: 9452300
- 56. Mimori T, Matsumoto T, Calvopiña MH, Gomez EA, Saya H, Katakura K, et al. Usefulness of sampling with cotton swab for PCR-diagnosis of cutaneous leishmaniasis in the New World. Acta Trop. 2002. https://doi.org/10.1016/s0001-706x(01)00215-7 PMID: 11835896
- 57. Gomes AHS, Armelin IM, Menon SZ, Pereira-Chioccola VL. Leishmania (V.) braziliensis: Detection by PCR in biopsies from patients with cutaneous leishmaniasis. Exp Parasitol. 2008. https://doi.org/10. 1016/j.exppara.2008.02.014 PMID: 18442815
- Yehia L, Adib-Houreih M, Raslan WF, Kibbi AG, Loya A, Firooz A, et al. Molecular diagnosis of cutaneous leishmaniasis and species identification: Analysis of 122 biopsies with varied parasite index. J Cutan Pathol. 2012. https://doi.org/10.1111/j.1600-0560.2011.01861.x PMID: 22335594

- 59. Saldanha MG, Queiroz A, Machado PRL, De Carvalho LP, Scott P, De Carvalho Filho EM, et al. Characterization of the histopathologic features in patients in the early and late phases of cutaneous leishmaniasis. Am J Trop Med Hyg. 2017. https://doi.org/10.4269/ajtmh.16-0539 PMID: 28115669
- Boni SM, Oyafuso LK, Soler R de C, Lindoso JAL. Efficiency of noninvasive sampling methods (swab) together with Polymerase Chain Reaction (PCR) for diagnosing american tegumentary leishmaniasis. Rev Inst Med Trop Sao Paulo. 2017. https://doi.org/10.1590/S1678-9946201759038 PMID: 28591266
- Boggild AK, Valencia BM, Veland N, Ramos AP, Calderon F, Arevalo J, et al. Non-invasive cytology brush pcr diagnostic testing in mucosal leishmaniasis: Superior performance to conventional biopsy with histopathology. PLoS One. 2011; 6: e26395–e26395. <u>https://doi.org/10.1371/journal.pone.</u> 0026395 PMID: 22046280
- Agrawal S, Khandelwal K, Bumb RA, Oghumu S, Salotra P, Satoskar AR. Short report: Pediatric Cutaneous leishmaniasis in an endemic region in India. Am J Trop Med Hyg. 2014. <u>https://doi.org/10.4269/ ajtmh.13-0737 PMID: 25294618</u>
- Aksoy M, Doni N, Ozkul HU, Yesilova Y, Ardic N, Yesilova A, et al. Pediatric Cutaneous Leishmaniasis in an Endemic Region in Turkey: A Retrospective Analysis of 8786 Cases during 1998–2014. PLoS Negl Trop Dis. 2016. https://doi.org/10.1371/journal.pntd.0004835 PMID: 27415630
- Uribe-Restrepo A, Cossio A, Desai MM, Dávalos D, Castro M del M. Interventions to treat cutaneous leishmaniasis in children: A systematic review. PLoS Negl Trop Dis. 2018. <u>https://doi.org/10.1371/journal.pntd.0006986</u> PMID: 30550538
- 65. Valencia BM, Veland N, Alba M, Adaui V, Arevalo J, Low DE, et al. Non-Invasive Cytology Brush PCR for the Diagnosis and Causative Species Identification of American Cutaneous Leishmaniasis in Peru. Mulvenna J, editor. PLoS One. 2012; 7: e49738. https://doi.org/10.1371/journal.pone.0049738 PMID: 23185421
- 66. Pereira L de OR, Moreira RB, de Oliveira MP, Reis S de O, de Oliveira Neto MP, Pirmez C. Is *Leishmania (Viannia) braziliensis* parasite load associated with disease pathogenesis? Int J Infect Dis. 2017; 57: 132–137. https://doi.org/10.1016/j.ijid.2017.01.036 PMID: 28167253
- Camacho E, Rastrojo A, Sanchiz Á, González-De La Fuente S, Aguado B, Requena JM. *Leishmania* Mitochondrial Genomes: Maxicircle Structure and Heterogeneity of Minicircles. 2019 [cited 11 Apr 2020]. https://doi.org/10.3390/genes10100758 PMID: 31561572
- 68. van der Auwera G, Dujardina JC. Species typing in dermal leishmaniasis. Clin Microbiol Rev. 2015; 28: 265–294. https://doi.org/10.1128/CMR.00104-14 PMID: 25672782
- Nicolas L, Milon G, Prina E. Rapid differentiation of Old World *Leishmania* species by LightCycler polymerase chain reaction and melting curve analysis. J Microbiol Methods. 2002; 51: 295–299. https://doi.org/10.1016/S0167-7012(02)00099-4 PMID: 12223289
- 70. Nath-Chowdhury M, Sangaralingam M, Bastien P, Ravel C, Pratlong F, Mendez J, et al. Real-time PCR using FRET technology for Old World cutaneous leishmaniasis species differentiation. Parasites and Vectors. 2016; 9: 255. https://doi.org/10.1186/s13071-016-1531-4 PMID: 27141967
- Kocher A, Valière S, Bañuls AL, Murienne J. High-throughput sequencing of kDNA amplicons for the analysis of *Leishmania* minicircles and identification of Neotropical species. Parasitology. 2018; 145. https://doi.org/10.1017/S0031182017002013 PMID: 29144208
- 72. Sevilha-Santos L, dos Santos Júnior ACM, Medeiros-Silva V, Bergmann JO, da Silva EF, Segato LF, et al. Accuracy of qPCR for quantifying *Leishmania* kDNA in different skin layers of patients with American tegumentary leishmaniasis. Clin Microbiol Infect. 2019. <u>https://doi.org/10.1016/j.cmi.2018.04.025</u> PMID: 29730222
- 73. Romero GAS, de Farias Guerra M V., Paes MG, Macedo V d. O. Comparison of Cutaneous Leishmaniasis Due to Leishmania (Viannia) braziliensis and L. (V.) guyanensis in Brazil: Clinical Findings and Diagnostic Approach. Clin Infect Dis. 2001; 32: 1304–1312. https://doi.org/10.1086/319990 PMID: 11303265
- 74. Arevalo J, Ramirez L, Adaui V, Zimic M, Tulliano G, Miranda-Verástegui C, et al. Influence of *Leishmania (Viannia)* Species on the Response to Antimonial Treatment in Patients with American Tegumentary Leishmaniasis. J Infect Dis. 2007; 195: 1846–1851. <u>https://doi.org/10.1086/518041</u> PMID: 17492601
- 75. Llanos-Cuentas A, Tulliano G, Araujo-Castillo R, Miranda-Verastegui C, Santamaria-Castrellon G, Ramirez L, et al. Clinical and Parasite Species Risk Factors for Pentavalent Antimonial Treatment Failure in Cutaneous Leishmaniasis in Peru. Clin Infect Dis. 2008; 46: 223–231. <u>https://doi.org/10.1086/</u> 524042 PMID: 18171254
- 76. Olliaro P, Vaillant M, Arana B, Grogl M, Modabber F, Magill A, et al. Methodology of Clinical Trials Aimed at Assessing Interventions for Cutaneous Leishmaniasis. PLoS Negl Trop Dis. 2013. <u>https://doi.org/10.1371/journal.pntd.0002130 PMID: 23556016</u>

- 77. Oré M, Sáenz E, Cabrera R, Sanchez JF, De Los Santos MB, Lucas CM, et al. Outbreak of cutaneous leishmaniasis in peruvian military personnel undertaking training activities in the amazon basin, 2010. Am J Trop Med Hyg. 2015; 93: 340–346. https://doi.org/10.4269/ajtmh.15-0107 PMID: 26078320
- 78. Valdivia HO, De Los Santos MB, Fernandez R, Baldeviano GC, Zorrilla VO, Vera H, et al. Natural Leishmania infection of *Lutzomyia auraensis* in Madre de Dios, Peru, detected by a fluorescence resonance energy transfer-based real-time polymerase chain reaction. Am J Trop Med Hyg. 2012; 87: 511–517. https://doi.org/10.4269/ajtmh.2012.11–0708
- 79. Zorrilla V, De Los Santos MB, Espada L, Santos R del P, Fernandez R, Urquia A, et al. Distribution and identification of sand flies naturally infected with *Leishmania* from the Southeastern Peruvian Amazon. PLoS Negl Trop Dis. 2017; 11: e0006029–e0006029. https://doi.org/10.1371/journal.pntd.0006029 PMID: 29107954
- Zorrilla V, Vásquez G, Espada L, Ramírez P. Vectores de la leishmaniasis tegumentaria y la Enfermedad de Carrión en el Perú: una actualización. Rev Peru Med Exp Salud Publica. 2017; 34: 485. <a href="https://doi.org/10.17843/rpmesp.2017.343.2398">https://doi.org/10.17843/rpmesp.2017.343.2398</a> PMID: 29267774
- De Los Santos MB, Ramirez IM, Rodriguez JE, Beerli P, Valdivia HO. Genetic diversity and population structure of *Leishmania (Viannia) braziliensis* in the Peruvian jungle. PLoS Negl Trop Dis. 2022; 16. https://doi.org/10.1371/journal.pntd.0010374 PMID: 35605021
- 82. Travi BL, De los Santos MB, Shelite TR, Santos RP, Rosales LA, Castellanos-Gonzalez A, et al. Diagnostic Efficacy of Recombinase-Polymerase-Amplification Coupled with Lateral Flow Strip Reading in Patients with Cutaneous Leishmaniasis from the Amazonas Rainforest of Perú. Vector-Borne Zoonotic Dis. 2021; 21. https://doi.org/10.1089/vbz.2021.0038 PMID: 34665665
- Grogl M, Joya CA, Saenz M, Quispe A, Rosales LA, Santos RDP, et al. Evaluation of a diagnostic device, CL Detect rapid test for the diagnosis of new world cutaneous leishmaniasis in Peru. PLoS Negl Trop Dis. 2023; 17. https://doi.org/10.1371/journal.pntd.0011054 PMID: 36913433